Author’s response to reviews

Title: Exercise prescription for patients with multiple sclerosis; Potential benefits and practical recommendations.

Authors:

Farzin Halabchi (fhalabchi@tums.ac.ir)
Zahra Alizadeh (z_alizadeh@tums.ac.ir)
Mohammad Ali Sahraian (msahrai@tums.ac.ir)
Maryam Abolhasani (m_abolhasani@tums.ac.ir)

Version: 2 Date: 18 May 2017

Author’s response to reviews:

Dear Editor:

Please find our revised article entitled: “Exercise prescription for patients with multiple sclerosis; Potential benefits and practical recommendations.”

In this manuscript, we aimed to determine the beneficial effects of the exercise training, on patients with multiple sclerosis.

Please consider the revised manuscript for publication in the BMC Neurology Journal. I accept the responsibility for ensuring that the submitted revised manuscript has the approval of all the co-authors. Moreover, I revised this manuscript according to the points of honorable reviewers for editing English grammar. I used track changes for editing the text.

Answers to reviewer 1:

Dear reviewer,

Thank you for valuable comments.

We tried to correct Grammatical errors in all parts of the text as much as possible.
Answers to reviewer 2:

Dear reviewer,

Thank you for valuable comments.

We added "narrative" before review at line 132 in introduction section.

We tried to correct Grammatical errors, typing mistakes and punctuations in all parts of the text, as well as in citations, according to your valuable comments as much as possible.

Thank you very much for your kind considerations.

Faithfully yours,

Maryam Abolhasani.

Corresponding author.